Author's response to reviews

Title: Comparison of STI-related consultations among ethnic groups in the Netherlands: an epidemiologic study using electronic records from the general practice

Authors:

Petra J Woestenberg (Petra.Woestenberg@rivm.nl)
Aloysia AM van Oeffelen (Louise.van.Oeffelen@rivm.nl)
Irina Stirbu-Wagner (irina_stirbu@yahoo.com)
Birgit HB van Benthem (Birgit.van.Benthem@rivm.nl)
Jan EAM van Bergen (JvanBergen@soaaids.nl)
Ingrid VF van den Broek (Ingrid.van.den.Broek@rivm.nl)

Version: 2
Date: 30 January 2015

Author's response to reviews:

Figure 2 was not uploaded correctly. We have changed the format of the figure to try it again.
We also like to suggest the following potential peer reviewers:

Sebastian Kalwij
sebastiankalwij@mac.com

Kristin Hendrickx
kristin.hendrickx@telenet.be
Department of General Practice, University of Antwerp, Belgium,
Universiteitsplein, 1, 2610, Wilrijk, Belgium,